Author’s response to reviews

Title: Psychoeducational Groups versus Waitlist in Treatment of Attention-Deficit Hyperactivity/Impulsivity Disorder (ADHD) in Adults: A Protocol for a Pilot Randomized Waitlist-Controlled Multicenter Trial.

Authors:

Jonas Vaag (jonas.vaag@nord.no)

Lara-Cabrera Mariela (Cabrera.Mariela.Loreto.Lara@stolav.no)

Odin Hjemdal (odin.hjemdal@svt.ntnu.no)

Bjørn Gjervan (bjorn.gjervan@hnt.no)

Terje Torgersen (terje.torgersen@stolav.no)

Version: 1 Date: 18 Nov 2018

Author’s response to reviews:

Dear editor of Pilot and Feasibility Studies.

On behalf of the authors, we thank you for the opportunity to resubmit this paper.

Thank you for sending us the comments on the manuscript. In an attached file, we address the issues raised and have incorporated the recommendations and changes into the paper’s text. We believe that the reviewers comments has improved the manuscript.

In brief, we acknowledge reviewer 1’s comments regarding the methodological issues and objectives of the trial. A more detailed description of the objectives and the conditions is added. We agree that the statistics were described as if it were an efficacy trial. They have now been changed in the manuscript.

Most of the suggestions made by reviewer 2 have been taken into consideration.

Following this letter are our responses to the reviewers, a manuscript marked with changes, as well as a clean manuscript that have undergone english proof editing.

We look forward to your re-evaluation of the manuscript.

On behalf of the authors.

Yours sincerely,